
Observational log CHARA/VEGA 2015-11-04
Observers: Fredéric, Orlagh, and Nic
Instruments: VEGA & CLIMB(tracking)

Configuration:
Telescope Beam PoP

E1 B3 P1
E2∗ B1 P2
W1 B3 P1
W2 B2 P5
S1 B2 P4
S2 B1 P1

∗E2 not available

Beginning of the observation:
01.00 Arrived at the control room.
01.00 Humidity still at 100%, Nic left mountain. We are still (too) optimistic, humidity

at W2 has dropped to 90% !
08.00 Humidity dropping slightly but still 90% for all telescopes. We are still a little

hopeful, W2 at about 85%!
10.00 Not so hopeful anymore.

End of Night

Time is in UT+00.00, blue.. science target, red.. calibrator, green.. spectral calibration, gold.. additional information.
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